Author's response to reviews

Title: Household's Perception of Climate Change and Human Health Risks: A community perspective

Authors:
Md. Aminul Haque (aminuldps@gmail.com)
Shelby Suzanne Yamamoto (shelby.yamamoto@urz.uni-heidelberg.de)
Ahmad Azam Malik (Malik@uni-heidelberg.de)
Rainer Sauerborn (Rainer.Sauerborn@urz.uni-heidelberg.de)

Version: 3 Date: 14 October 2011

The Environmental Health Editorial Team
Re: MSMS: 1872216238587474 Household's Perception of Climate Change and Human Health Risks: A community perspective.

Md. Aminul Haque, Shelby Suzanne Yamamoto, Ahmad Azam Malik and Rainer Sauerborn

Thank you for your e-mail on September 26, 2011, regarding the reviews for the above manuscript. We appreciate the reviewers’ comments, as well as your guidance for revising the manuscript. We have attached a “track changes” version that includes corrections made in response to your instructions and to the reviewers’ comments. Below is a point-by-point response to the reviewers’ comments, including descriptions of revisions made to the manuscript. We feel that we have addressed all of the comments and hope that you will consider the manuscript acceptable for publication in your journal.
Correspondence about this manuscript should be directed to:
Md. Aminul Haque (aminuldp@gmail.com)